 
CENWP-OP-B                                                                                               16 January, 2001

MEMORANDUM FOR CENWP-OP, ATTN: Gary Johnson

SUBJECT: Bonneville Lock and Dam, Fishway and Fish Activities for week 02 of 2000, which covers the period 07 through 13 January.
1.  OPERATION SUMMARY.

1.1 Average daily river flow ranged from 126.9 to 137.9 kcfs.  Average daily tailwater elevation for the project ranged from 12.1’ to 12.6’ msl.  Average daily forebay elevation for the project ranged from 73.1’ to 76.4’ msl.  River temperature stayed at 44° F all week.  Spill bay 1, which supplies attraction flow for Cascades Island ladder, remained open 4”.  Spill bay 18, which supplies attraction flow for Bradford Island ladder, was opened to 4”.

1.2 Unit 3 remained out of service for turbine rehab.   Unit 14 was out of service from 0400 until 1800 on 08 January for orifice ring replacement.  Unit 16 was out of service from 2400 08 January until 1306 12 January for servomotor repair.  Unit 17 was out of service from 2400 08 January until 1630 09 January for orifice ring replacement.  Unit 18 was out of service from 0400 until 1642 10 January for orifice ring replacement.

1.3 Fish Unit 2 remains out of service for annual maintenance.  Fish Unit 1 went out of service at 1900 08 January for annual maintenance.  

2.  ACTIVITIES. 
2.1 AWS Closures.  PH2 Ice and Trash (used as AWS back up) was closed at 1915 08 January.  FV6-9 was closed for the Washington Shore dewatering at 2045 08 January.

2.2 STS Inspections.  STS’s are pulled from all units.

2.3 Dewatering and Fish Salvage.  Washington Shore fishway was dewatered on 09 January.  All fish were guided to tailwater with the exception of fish in the UMT.  Approximately 42 adult steelhead were removed from the UMT, transported by tank truck, then released upstream of the navigation lock.  All fish were in good condition upon release.

 
3.   RESEARCH. 

3.1 NMFS Lamprey Tagging and Radio Tracking.  NMFS completed trapping and tagging lamprey in the AFC&MF.  They are still tracking the fish back through the project in cooperation with the University of Idaho.

3.2 WDFW Fish Counting.  WDFW continues 24-h/d video counting at Cascade Island and Bradford Island.

4.  FISHWAYS. 
4.1 Project biologists inspected on 08, 10, and 11 January this reporting week.

4.2 Fish Passage Plan Observations.

DATE
LOCATion
FPP Deviation
CAUSE
RESPONSE

All week
Cascades Island diffuser FG6-18.
Diffuser is stuck partially open and should be closed.
Unknown.
Will repair during winter maintenance period.

All week.
PH1 collection channel electronic velocity meter.
Meter was not working and should register 1.5-4.0 fps per the FPP.
Unknown.
Investigating alternative method of measure.  



All week
Cascade Island bird wires.
One wire is down.
Rangers cut wire to remove entangled gull.
Will repair during winter maintenance period.

All week
Chain Gate 7A
Gate was closed.  Should be set to 72’ msl.
Riggers are unable to remove the bulkhead in the slot.
Waiting for the proper equipment to remove bulkhead.

01/08
DSM2 channel elevation.
Water level was below 64.2’ msl.
Riggers were changing out the 13” orifice rings with 12 11/16”.
No action taken.

4.3  Fish Ladder Temperatures.  Readings are taken 3 weirs down from the count station at Bradford Island and at the junction of the UMT and the main ladder at Washington shore.  Thermometers are currently undergoing calibration.
4.4 Sensor Calibration.  Water level sensors were checked for calibration on 08 January this reporting week. 

4.5 Zebra Mussel Monitoring.  The zebra mussel monitoring station was inspected and showed no signs of colonization this reporting week.

5. MISCELLANEOUS

5.1 Because of mechanical reasons, DSM2 screen cleaners are run once per shift (or more often depending on debris loads) to reduce wear and tear on the equipment. 

5.2 Bradford Island fishways were watered up at 1115 08 January.
5.3 Cascades Island exit was opened to allow fish passage while Washington shore is out of service.  
5.4 Washington Shore went out of service at 1000 09 January.
5.5 PH1 Collection Channel was inspected by divers on 11 January.
5.6 DSM1 remains out of service for maintenance.
                                                                                                     DEBORAH L. CHENOWETH

                                 

                                                              Operations Project Manager
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